MASSACHVSETTS INSTITVTE OF TECHNOLOGY Department of Electrical Engineering and Computer Science 6.090—Building Programming Experience IAP, 2006

Quiz I

NAME:

Section Number:

Tutor’s Name:

PART Value Grade Grader

1 18 2 20 3 30 4 12 5 20 Total 100

If you wish to write a note on any problem, put a check in the box next to the problem number, and use the space provided at the end of the answer sheet. Notes written outside the answer boxes on this page will be ignored.

PART 0 (0 points)

How many hours of sleep did you get last night?

PART 1 (18 points)

Q1.(4)

Q2.(2) Q3.(4)

Q4.(2) Q5.(4)

Q6.(2)

PART 2 (20 points) Q7.(20)

PART 3 (30 points) Q8.(10)

Q9.(10)

Q10.(5)

Q11.(5)

PART 4 (12 points)

Q12.(3)

Q13.(3)

Q14.(3)

Q15.(3)

PART 5 (20 points)

Q16.(2)

Q17.(2)

Q18.(2)

Q19.(2)

Q20.(2)

Q21.(2)

Q22.(2)

Q23.(2)

Q24.(2)

Q25.(2)

The rest of this page is for your use in marking special notes (see instructions on front page).